Author's response to reviews

Title: Views and experiences of health care professionals towards the use of African traditional, complementary and alternative medicines among patients with HIV infection: the case of eThekwini health district, South Africa.

Authors:

Manimbulu Nlooto (Nlooto@ukzn.ac.za)

Version: 8 Date: 11 May 2015

Author's response to reviews:

Author's response to reviews (May 2015)

Title: Views and experiences of healthcare providers towards the use of African traditional, complementary and alternative medicines among patients with HIV infection: the case of eThekwini health district, South Africa

Author: Manimbulu Nlooto (Nlooto@ukzn.ac.za)

Version 6: date 11 May 2015

Author's response to reviews: see over

Durban 11 May 2015

The Biomed Central Editorial Team

Object: MS: 6639307221467707: Views and experiences of healthcare providers towards the use of African traditional, complementary and alternative medicines among patients with HIV infection: the case of eThekwini health district, South Africa
Manimbulu Nlooto

Thank you for consideration of my manuscript for publication in your journal. Please find below my responses to the editor’s request.

Comments from the editor:

Copyediting should be done to improve the style of written English.

• Response

1. The manuscript has been edited by a professional English language editing service with expertise in health sciences.

2. The structure of the manuscript main text as well as the abstract has been improved to read well for a general and specific readership.

3. As much as possible the manuscript text complies with BMC-series medical journals-authors’ checklist for manuscript formatting.